Author's response to reviews

Title: Vegan diet and blood lipid profiles: a cross-sectional study of pre- and postmenopausal women

Authors:

Yee-Wen Huang (47445@cch.org.tw)
Zhi-Hong Jian (s841039@yahoo.com.tw)
Hui-Chin Chang (cshe215@csh.org.tw)
Oswald Ndi Nfor (nforoswald2@yahoo.com)
Pei-Chieh Ko (ko52070@hotmail.com)
Chia-Chi Lung (dinoljc@csmu.edu.tw)
Long-Yau Lin (xillin681113@gmail.com)
Chien-Chang Ho (ccho1980@gmail.com)
Yi-Chen Chiang (jiy@csmu.edu.tw)
Yung-Po Liaw (Liawyp@csmu.edu.tw)

Version: 5
Date: 20 March 2014

Author's response to reviews: see over
Dear Sir:

Thank you for sending me the editorial comments on the manuscript (No. MS: 1839581692116316) entitled “Vegan diet and blood lipid profiles: a cross-sectional study of pre and postmenopausal women”. Authors have had the language in the manuscript edited by a native-English speaker with scientific expertise. We hope that the revised manuscript is now acceptable for publication in BMC Women’s Health.

Should there be any future correspondences concerning this resubmission, please do not hesitate to contact me. Thank you again for your kindly help and I look forward to hearing from you soon.

Best regards,

Sincerely,

Yung-Po Liaw
Department of Public Health and Institute of Public Health,
Chung Shan Medical University,
No. 110 Sec 1 Chien-Kuo N. Road, Taichung City 40201, Taiwan
R.O.C
Phone: +88-642-4730022 ext.11838
Fax: +88-642-3248179
Email: Liawyp@csmu.edu.tw
Point-to-point response to the editorial comments

Copyediting: We recommend that you copyedit the paper to improve the style of written English.

Response:
Thanks for your great comments. Authors have had the language in the manuscript edited by a native-English speaker with scientific expertise.